Author’s response to reviews

Title: Ethnic disparities in children's oral health: findings from a population-based survey of grade 1 and 2 schoolchildren in Alberta, Canada

Authors:

Congshi Shi (congshi.shi@ucalgary.ca)

Peter Faris (peter.faris@ahs.ca)

Deborah McNeil (debbie.mcneil@ahs.ca)

Steven Patterson (steven.patterson@ualberta.ca)

Melissa Potestio (melissa.potestio@ahs.ca)

Salima Thawer (sthawer@ucalgary.ca)

Lindsay McLaren (lmclaren@ucalgary.ca)

Version: 1 Date: 17 Jan 2017

Author’s response to reviews:

Please see our response to reviewer document uploaded later in this application.